\documentclass[12pt]{amsart}

\setlength{\parindent}{0pt}
\setlength{\textheight}{9in}
\pagestyle{empty}

\newcommand{\answerspace}{\vspace*{1in}}

\begin{document}

\thispagestyle{empty}

\textbf{Math 126: Calculus II}
\hfill\makebox[2.5in][r]{Name:\hrulefill}
 
\bigskip
\begin{center}
\textbf{Extra Credit Assignment}
\end{center}
\bigskip

We would like your feedback to help us improve the way we use Mathematica
in the course in the future.  You will receive ten extra credit assignment
points if you answer the following questions thoughtfully.

\bigskip

1. Where did you use Mathematica?

\answerspace

2. Describe the main technical problems you experienced using Mathematica
(network connectivity, memory shortages, printer problems, crashes,
etc.) and how often they occurred.

\answerspace

3. How long did it typically take you to complete the Mathematica portion of
an assignment?

\answerspace

4. Do you think learning to use Mathematica was worthwhile either
in the context of this course or more generally?

\answerspace

5. What suggestions do you have for how we can improve the use of
Mathematica in the course in the future?

\answerspace

\end{document}

